January 6, 2024

Dr.

ELENA DANIELA SERBAN
Editor-in-Chief
World Journal of Clinical Pediatrics

Thank you for taking into consideration the academic work entitled "Sociodemographic determinants associated with breastfeeding in term newborns with low birth weight in Latin American countries" for possible publication in "World Journal of Clinical Pediatrics".

The following is the list of changes made to the manuscript.

a. RESPONSE TO REVIEWERS: MINOR LANGUAGE POLISHING

Linguistic revision of the following paragraphs (yellow in the text)

Core tip: Last paragraph: Sentence structure clarified and simplified.

Introduction:

Paragraph 1. sentence 3.

Paragraph 2. Sentence 2.


Complete paragraph 7.

Methods

Paragraph 5. Sentence 1.

Paragraph 7. Sentence 2
Results

Discussion

2. Abbreviations

Abbreviations defined in their first order of appearance.

Table 2

The meaning of the abbreviations in the table is included.

b. EDITORIAL OFFICE’S COMMENTS

The reference in Figure 1 is cited and delivered in PPT format.

PMID and DOI are included in all references.

Highlights are integrated after conclusions and before the references section.

Abstract is modified according to the editor's comments.
Best Regards

CARLOS JAVIER AVENDAÑO VÁSQUEZ

E-mail: cjavasquez@uan.edu.co